\nocite{Ottolini.sep.14.281}
\nocite{Ottolini.sep.15.97}
\nocite{Bloxsom.sep.15.251}
\nocite{Ottolini.sep.15.283}
\nocite{McMechan.sep.20.25}
\nocite{Ottolini.sep.20.97}
\nocite{Ottolini.sep.20.117}
\nocite{Ottolini.sep.25.63}
\nocite{Ottolini.sep.25.363}
\nocite{Ottolini.sep.26.83}
\nocite{Ottolini.sep.26.119}
\nocite{Ottolini.sep.26.297}
\nocite{Ottolini.sep.26.299}
\nocite{Ottolini.sep.26.329}
\nocite{Ottolini.sep.26.337}
\nocite{Ottolini.sep.28.5}
\nocite{Ottolini.sep.28.121}
\nocite{Ottolini.sep.28.139}
\nocite{Ottolini.sep.28.189}
\nocite{Ottolini.sep.28.321}
\nocite{Ottolini.sep.28.335}
\nocite{Ottolini.sep.30.43}
\nocite{Ottolini.sep.30.201}
\nocite{Ottolini.sep.32.15}
\nocite{Claerbout.sep.32.37}
\nocite{Ottolini.sep.32.117}
\nocite{Ottolini.sep.32.183}
\nocite{Ottolini.sep.32.205}
\nocite{Ottolini.sep.33.00}
\nocite{Ottolini.sep.33.15}
\nocite{Ottolini.sep.33.35}
\nocite{Ottolini.sep.33.46}
\nocite{Ottolini.sep.33.64}
\nocite{Ottolini.sep.33.69}
\nocite{Ottolini.sep.33.73}
\nocite{Ottolini.sep.33.74}
\nocite{Ottolini.sep.33.79}
\nocite{Ottolini.sep.33.85}
\nocite{Ottolini.sep.33.88}
\nocite{Ottolini.sep.33.90}
\nocite{Ottolini.sepphd.33}
\nocite{Ottolini.sep.33}
\nocite{Ottolini.sep.35.235}
\nocite{Ottolini.sep.35.241}
\nocite{Ottolini.sep.35.302}
\nocite{Ottolini.sep.35.304}
\nocite{Ottolini.sep.35.342}
\nocite{Ottolini.sep.35.352}
\nocite{Ottolini.sep.37.59}
\nocite{Ottolini.sep.37.143}
\nocite{Ottolini.sep.37.179}
\nocite{Ottolini.sep.37.181}
\nocite{Ottolini.sep.37.187}
\nocite{Li.sep.38.141}
\nocite{Ottolini.sep.38.171}
\nocite{Ottolini.sep.48.295}
\nocite{Ottolini.sep.48.359}
\nocite{Ottolini.sep.50.35}
\nocite{Ottolini.sep.50.279}
\nocite{Ottolini.sep.50.291}
\nocite{Ottolini.sep.50.297}
\nocite{Ottolini.sep.51.351}
\nocite{Ottolini.sep.51.363}
\nocite{Ottolini.sep.51.367}
\nocite{Ottolini.sep.51.397}
\nocite{Ottolini.sep.51.409}
\nocite{Ottolini.sep.51.415}
\nocite{Ottolini.sep.51.421}
\nocite{Ottolini.sep.56.203}
\nocite{Ottolini.sep.56.273}
\nocite{Ottolini.sep.57.457}
\nocite{Ottolini.sep.57.559}
\nocite{Ottolini.sep.57.579}
\nocite{Ottolini.sep.Index.0}
\nocite{Ottolini.sep.Index.1}
\nocite{Ottolini.sep.Index.3}
\nocite{Ottolini.sep.Index.9}
\nocite{Ottolini.sep.Index.13}
\nocite{Ottolini.sep.Index.27}
\nocite{Ottolini.sep.Index.43}
\nocite{Ottolini.sep.Index.101}
\nocite{Nichols.sep.59.73}
\nocite{Ottolini.sep.59.209}
\nocite{Ottolini.sep.59.255}
\nocite{Ottolini.sep.59.269}
\nocite{Ottolini.sep.59.281}
\nocite{Ottolini.sep.60.413}
\nocite{Ottolini.sep.60.419}
\nocite{Ottolini.sep.60.433}
\nocite{Ottolini.sep.61.319}
\nocite{Ottolini.sep.61.337}
\nocite{Ottolini.sep.65.247}
\nocite{Ottolini.sep.65.295}
\nocite{Ottolini.sep.65.315}
\nocite{Ottolini.sep.67.51}
\nocite{Ottolini.sep.67.167}


\title{Report bibliography}
This is my bibliography of articles
in the progress reports of the Stanford Exploration Project
written from 1974 to the present.
There are  95 
articles.
{\footnotesize
\putbib[SEP]
}
